Author's response to reviews

Title: Predictors of incident herpes simplex virus type 2 infections in young women at risk for unintended pregnancy in San Francisco

Authors:

- Nicholas J. Moss (njmoss@u.washington.edu)
- Cynthia C. Harper (harperc@obgyn.ucsf.edu)
- Katherine Ahrens (katherine.ahrens@sfdph.org)
- Katherine Scott (kate.scott@sfdph.org)
- Susan Kao (kao.susan@gmail.com)
- Nancy Padian (npadian@globalhealth.ucsf.edu)
- Tina Raine (irainet@obgyn.ucsf.edu)
- Jeffrey D. Klausner (jeff.klausner@sfdph.org)

Version: 2 Date: 7 June 2007

Author's response to reviews:

I am correctly problems encountered during a previous upload of the manuscript and tables.